Reference Texts for AOS 571 (on reserve in Geosciences Library , Fine Hall)

Batchelor, G.K. Introduction to Fluid Dynamics Cushman-Roisin, B. Introduction to Geophysical Fluid Dynamics* Gill, A.E. Atmosphere-Ocean Dynamics* Holton, J.R. Introduction to Dynamic Meteorology Kundu, P. Fluid Mechanics Landau, L.D. and Lifshitz, Fluid Mechanics Lindzen, R.S. Dynamics in Atmospheric Physics: Lecture Notes for an Intro Pedlosky, J. Geophysical Fluid Dynamics Peixoto, J.P. Physics of Climate Salmon, R. Lectures on Geophysical Fluid Dynamics

* should be available in University book store